Author's response to reviews

Title: Occupational Therapy in HomEcare Re-ablement Services (OTHERS): Protocol for a Randomised Controlled Trial

Authors:

Phillip J Whitehead (phillip.j.whitehead@nottingham.ac.uk)
Avril ER Drummond (avril.drummond@nottingham.ac.uk)
Marion F Walker (marion.walker@nottingham.ac.uk)
Ruth H Parry (ruth.parry@nottingham.ac.uk)
Ian McGeorge (imcgeorge@hotmail.co.uk)
Zaid Latif (zaid.latif@nottinghamcity.gov.uk)

Version: 2
Date: 22 April 2014

Author's response to reviews:

1. I have included the date the trial was registered within the abstract.
2. I have added a statement on informed consent to the abstract section and an explanation of the consent process is detailed within the main section of the manuscript.
3. I have included each author's initials within the contribution section.
4. I have added a figure title at the end of the main manuscript.